UC Davis UC Davis Previously Published Works

Title

Patient-reported symptoms during and after direct-acting antiviral therapies for chronic hepatitis C: The PROP UP study

Permalink https://escholarship.org/uc/item/7jc1f9bn

Journal Journal of Hepatology, 71(3)

ISSN 0168-8278

Authors

Evon, Donna M Sarkar, Souvik Amador, Jipcy <u>et al.</u>

Publication Date 2019-09-01

DOI

10.1016/j.jhep.2019.04.016

Peer reviewed



HHS Public Access

Author manuscript *J Hepatol.* Author manuscript; available in PMC 2020 September 01.

Published in final edited form as:

J Hepatol. 2019 September ; 71(3): 486–497. doi:10.1016/j.jhep.2019.04.016.

Patient-reported symptoms during and after direct acting antiviral therapies for chronic hepatitis C: The PROP UP Study

Donna M. Evon^{1,*}, Souvik Sarkar^{2,*}, Jipcy Amador³, Anna S. Lok⁴, Richard K. Sterling⁵, Paul W. Stewart³, Bryce B. Reeve⁶, Marina Serper⁷, Nancy Reau⁸, K. Rajender Reddy⁷, Adrian M. Di Bisceglie⁹, David R. Nelson¹⁰, Carol E. Golin¹¹, Joseph K. Lim¹², Michael W. Fried¹

¹Division of Gastroenterology and Hepatology, Department of Medicine, University of North Carolina, Chapel Hill, NC;

²Division of Gastroenterology and Hepatology, Department of Medicine, University of California at Davis, Davis, CA;

³Department of Biostatistics, University of North Carolina, Chapel Hill, NC;

⁴Division of Gastroenterology and Hepatology, Department of Internal Medicine, University of Michigan, Ann Arbor, MI;

⁵Division of Gastroenterology, Hepatology & Nutrition, Department of Internal Medicine, Virginia Commonwealth University, Richmond, VA;

Declaration of interest: Donna M. Evon receives research funding from Gilead and Merck. Michael Fried has received research funding from and served as a consultant for AbbVie, BMS, Gilead, and Merck, and TARGET PharmaSolutions. Stock in TARGET PharmaSolutions is held in an independently managed trust. Anna S. Lok has received research support from BMS, Gilead, TARGET PharmaSolutions, AbbVie (ended in 2016), and Merck (ended in 2016); and served as an advisor for Gilead (interrupted between 1/2016–12/2018). Richard K. Sterling has received research support from AbbVie, BMS, Gilead, Merck, and Roche and served as a consultant for Merck, Bayer, Salix, AbbVie, Gilead, Jansen, ViiV, Baxter, and Pfizer. Joseph K. Lim has received research support (paid to Yale University) and served as a consultant for Bristol-Myers Squibb and Gilead. Nancy Reau has received research funding (paid to Rush) from AbbVie and Intercept and has served as a consultant for Merck, AbbVie, AbbVie, Abbott, and Gilead. Souvik Sarkar served on a Gilead and Abbvie Advisory Board and received grant support from Gilead through UCSF (paid to UC Davis). David R. Nelson has received research grant support from AbbVie, BMS, Gilead, Janssen, and Merck and owns stock in TARGET PharmaSolutions. K. Rajender Reddy is an Ad-Hoc Advisor to Gilead, BMS, Janssen, Merck, Abbvie, Shionogi, and Dova and has received research support from Gilead, BMS, Janssen, Merck, AbbVie, Intercept, Mallinckrodt, and Conatus (paid to the University of Pennsylvania). Adrian M. Di Bisceglie has received research support from AbbVie, BMS and Gilead and has served on advisory boards for AbbVie, BMS, Gilead and Merck. He serves as Chair of the Steering Committee for TARGET HCC, a registry study funded by TARGET PharmaSolutions. Paul Stewart has served as a consultant to TARGET PharmaSolutions. Jipcy Amador served as a biostatistics intern at TARGET PharmaSolutions in 2017. Carol E. Golin, Bryce Reeve, and Marina Serper declare that they have no conflict of interests to disclose.

Clinicaltrial.gov:

Corresponding Author: Donna M. Evon, PhD, Associate Professor, Director of Psychosocial Services and Research, Division of Gastroenterology and Hepatology, University of North Carolina, Chapel Hill, NC 27599, Telephone: (919) 966-6732, Donna_evon@med.unc.edu.

Author contributions: DME: study concept and design; analysis and interpretation of data; drafting of the manuscript; critical revision of the manuscript for important intellectual content; SS: manuscript drafting, data review, patient recruitment; JA: Data retrieval and editing, statistical computations and data analysis; graphical development; PWS: study design, analysis strategy, statistical computations, manuscript preparation and revision; ASL, RKS, NR: data review, critical review of manuscript, patient recruitment; data collection; BBR, CEG: study design, critical revision of the manuscript, manuscript revisions; MS, KRR, JKL, DRN, ADB, MWF: patient recruitment, data collection, manuscript revisions.

^c connotes co-first authorship

Publisher's Disclaimer: This is a PDF file of an unedited manuscript that has been accepted for publication. As a service to our customers we are providing this early version of the manuscript. The manuscript will undergo copyediting, typesetting, and review of the resulting proof before it is published in its final form. Please note that during the production process errors may be discovered which could affect the content, and all legal disclaimers that apply to the journal pertain.

⁶Department of Population Health Sciences, Duke University, Durham, NC;

⁷Division of Gastroenterology and Hepatology, Department of Medicine, University of Pennsylvania, Philadelphia, PA;

⁸Department of Internal Medicine, Section of Hepatology, Rush University, Chicago, IL;

⁹Division of Gastroenterology and Hepatology, Department of Internal Medicine, Saint Louis University, St. Louis, MO;

¹⁰Division of Gastroenterology, Hepatology & Nutrition, Department of Medicine, University of Florida, Gainesville, FL;

¹¹Division of General Medicine and Clinical Epidemiology, Department of Medicine, Department of Health Behaviors, University of North Carolina, Chapel Hill, NC,

¹²Digestive Diseases, Department of Internal Medicine, Yale University, New Haven, CT

Abstract

Background & Aims: A comprehensive analysis of changes in symptoms and functioning during and after direct acting antiviral (DAA) therapy for chronic hepatitis C virus (HCV) has not been conducted for patients treated in real-world clinical settings. We evaluated patient-reported outcomes (PROs) in a diverse cohort of HCV patients treated with commonly-prescribed DAAs.

Methods: PROP UP is a U.S. multicenter observational study of 1,601 HCV patients treated with DAAs in 2016–2017. PRO data were collected at baseline (T1), early on-treatment (T2), late on-treatment (T3) and 3-months post-treatment (T4). PRO mean change scores were calculated from baseline and a minimally important change (MIC) threshold was set at 5%. Regression analyses investigated patient and treatment characteristics independently associated with PRO changes on-treatment and post-treatment.

Results: Of 1,564 patients, 55% were male, 39% non-white, 47% had cirrhosis. Sofosbuvir/ ledipasvir was prescribed to 63%, sofosbuvir/velpatasvir to 21%, grazoprevir/elbasvir to 11%, and paritaprevir/ombitasvir/ritonavir+dasabuvir to 5%. During DAA therapy, mean PRO scores improved slightly in the overall cohort, but did not reach the 5% MIC threshold. Between 21–53% of patients experienced >5% improved PROs while 23–36% experienced >5% worse symptoms. Of 1,410 patients with evaluable sustained virologic response (SVR) data, 95% achieved SVR. Among those with SVR, all mean PRO scores improved, with the 5% MIC threshold met for fatigue, sleep disturbance, and functioning well-being. Regression analyses identified subgroups, defined by age 35–55, baseline mental health issues and a high number of health comorbidities as predictors of PRO improvements.

Conclusions: In real-world clinical practices, we observed heterogeneous patient experiences during and after DAA treatment. Symptom improvements were more pronounced in younger patients, those with baseline mental health issues and multiple comorbidities.

Lay Summary: Patients who received direct-acting antiviral medications for hepatitis C in usual care at several liver centers in the US on the whole did not experience significant changes in baseline symptoms during treatment. We observed a full range of patient experiences with some patients experiencing substantial symptom improvements, yet others experiencing less

improvements and some even worsening of symptoms. The 1346 patients who were cured of hepatitis C experienced improvements in fatigue, sleep disturbance, and functional well-being, and trends for improved pain and depression; whereas the 64 who were not cured experienced minimal improvements.

Keywords

HCV; liver; PRO; patient-reported outcome; quality of life; treatment; sleep; pain; functioning; symptom

Introduction

Patients with chronic hepatitis C virus (HCV) infection often report neuropsychiatric, somatic, and gastrointestinal symptoms including fatigue, sleep disturbance, musculoskeletal pain, depression, and abdominal pain (1–3). Patients may attribute these symptoms to HCV, a chronic viral infection associated with several extrahepatic disorders. Recent studies show that health-related quality of life and other patient-reported outcomes (PROs) improve during all-oral direct-acting antiviral (DAA) therapy and after patients achieve a sustained virological response (SVR) (4–6). These studies were based exclusively on data derived from industry-sponsored registration trials. It remains critical to determine if these findings can be generalized to patients treated in real-world clinical practices given inherent biases of registration trial data (7, 8).

Clinical trials enroll highly selected patients and typically under-represent important subgroups of the HCV population (9–11). Patients with psychosocial vulnerabilities (e.g., active psychiatric, drug use, alcohol abuse) are often excluded, yet these patients make up a sizeable majority of the population in need of treatment. In addition, a majority of these trials are comprised of predominantly White patients (66–97%) and those without advanced fibrosis (<20% cirrhosis) (9–11). Prior studies have also focused heavily on quality of life, work productivity, and fatigue outcomes but have not comprehensively evaluated specific somatic, gastrointestinal and neuropsychiatric symptoms often associated with chronic HCV (2). A more comprehensive description of symptom and function changes would enhance our understanding of the full spectrum of patients' experiences. Finally, PRO studies that allow for comparing patient experiences across different DAA regimens are lacking.

The current study enrolled a diverse cohort of patients initiating DAA therapy at several academic and community-based practices and includes a significant number of previously under-represented subgroups. We evaluated changes in overall symptom burden, specific HCV symptoms, functional well-being and health comorbidities in patients prescribed one of four DAA regimens: sofosbuvir/ledipasvir (SOF/LED); SOF/velpatasvir (SOF/VEL); Grazoprevir/Elbasvir(GRZ/ELB) and ombitasvir/paritaprevir/ritonavir with dasabuvir (PrOD) and provide a comprehensive characterization of real-world patient experiences during and after DAA therapy.

Patients and Methods

Study Design

The PROP UP study is funded by the Patient-Centered Outcomes Research Institute (PCORI) and described in detail in prior publications(12, 13). In brief, it is a multi-center, prospective, observational cohort study that enrolled patients across the U.S. to characterize patients' experiences associated with HCV, DAA treatments, and virologic cure. The current analysis utilized data collected at four time points: (T1) "Baseline" prior to starting DAA; (T2) "Early On-Treatment" week 4±2 weeks; (T3) "Late On-Treatment" last 2–3 weeks of therapy (e.g., weeks 10–12 for 12-week course); and (T4) "Post-Treatment" 12±2 weeks post-treatment.

Participants and Settings

PROP UP enrolled a total of 1,601 patients between January 2016 and October 2017 at 11 U.S. centers (9 academic hepatology centers, 2 private gastroenterology practices). All sites obtained Institutional Review Board approval and all patients provided informed consent prior to data collection.

Baseline characteristics

Sociodemographics.—Patients self-reported the following characteristics at baseline: date of birth, biological sex, race, ethnicity, educational attainment, annual household income, and employment status.

Mental health issues.—Patients who self-reported a history of any past psychiatric hospitalization or were taking psychiatric medications for 'depression, anxiety or nerve problems' at baseline were categorized as having mental health issues.

Drug use.—Patients who self-reported use of non-prescription illicit street drugs or misuse of prescription medications in the year before enrollment using validated items from the Substance Abuse Mental Illness Symptoms Screener (SAMISS) were categorized as having substance use issues (14).

Alcohol abuse.—Patients who scored 5 on three alcohol questions at baseline related to current frequency, quantity, and binge drinking using validated items from the Alcohol Use Disorders Identification Test (AUDIT) and the SAMISS were classified as having alcohol abuse (14, 15).

Cirrhosis.—Patients were classified as having cirrhosis (Yes/No) based on review of clinical, laboratory, imaging, histology, and transient elastography data in electronic health records. Aspartate aminotransferase (AST) to platelet ratio index (APRI) >2.0 and the Model for End-Stage Liver Disease (MELD) 12 were used to indicate advanced liver disease only in patients classified as having cirrhosis (16, 17). Adjudication of cases with inconsistent data was made by an experienced hepatologist (M.W.F.) or site investigators/ hepatologists.

Additional laboratory and treatment markers.—HCV genotype, HCV RNA level, AST, alanine aminotransferase (ALT), albumin, total bilirubin, platelets, hemoglobin, creatinine, international normalized ratio (INR), HIV, treatment regimen, treatment duration, and treatment experience were also recorded.

Sustained virologic response (SVR).—SVR was defined as an undetectable HCV RNA at 10 or more weeks after treatment completion. In 15 patients, lack of SVR was based on quantifiable HCV RNA around follow-up week 4.

Patient-Reported Outcomes

Additional details about these PROs are provided in the published protocol and baseline cohort analysis (12, 18).

Individual symptom clusters.—The National Institutes of Health's Patient-Reported Outcomes Measurement Information System® (PROMIS®) instruments were used to assess 10 specific symptoms that fall into three symptom clusters: *Neuropsychiatric Cluster* (depression, anxiety, anger, cognitive concerns); *Somatic Cluster* (pain interference, fatigue, sleep disturbance); and *Gastrointestinal Cluster* (abdominal pain, diarrhea, nausea/vomiting) (19–21). There is no PROMIS instrument to measure headache, therefore the Headache Impact Test (HIT-6) was used to capture headaches that may be associated with DAA therapy (22, 23). The HIT-6 has a 5-point Likert response scale ranging from "Never" to "Always." We previously evaluated the psychometric properties of the PROMIS and HIT-6 instruments in patients with HCV and have found satisfactory reliability and validity (18, 24). Higher PROMIS and HIT-6 scores reflect worse symptom experiences.

Overall symptom burden.—A comprehensive list of 32 symptoms common to many health conditions were assessed using the Memorial Symptom Assessment Scale (MSAS) (25, 26). Participants reported the presence or absence of symptoms (yes=1/no=0), and if present, its severity (0–4), frequency (0–4) and level of distress (0–4). The total score (TMSAS) could range from 0–4 and was multiplied by 10 for ease of interpretation with other PROs.

Functional well-being.—The HCV-PRO is a newly developed HCV-specific survey designed to evaluate the functioning and psychological well-being of patients with HCV (27, 28). The scale includes 16 items that measure various aspects of physical and emotional functioning, productivity, intimacy, and perceived quality of life related to having HCV. The 16 items are summed to produce a total score transformed on a scale of 0 (worst) to 100 (best). Unlike the other PROs, higher scores on the HCV-PRO are associated with *better* functional well-being. To display change in the HCV-PRO score in the same figures (below), the HCV-PRO change score was reverse coded.

Statistical Analysis Strategy

Primary Analysis.—The mean changes in each PRO score from T1 to T2, T3 and T4 Post-Treatment were estimated along with corresponding 95% confidence intervals (95% CIs). For the investigation of change from baseline to on-treatment in the primary analyses,

we used the average of the T2 and T3 PRO scores because these scores were moderately to highly correlated (r = 0.52 - 0.79). The estimates of mean changes were computed without adjustment for other covariates or confounders. The primary analysis focused on symptom-specific estimates of the magnitude of mean change from T1 to T2/T3 and T4 in the combined cohort, each of four DAA cohorts, those who achieved SVR and other subgroups of interest.

Descriptives.—Tabular and graphical methods were used to visualize the data (means, standard deviations (SD), range), change in PROs, or percent change from baseline.

Minimally Important Change (MIC).-To aid our clinical interpretation of the unadjusted PRO change scores, we defined a 5% change from baseline as the "minimally important change (MIC)" threshold based on the published literature and feedback from our patient engagement group about what amount of change would be meaningful to them related to treatment decision-making (5, 29). For the PROMIS symptom measures in which T-scores are standardized to a mean of 50 and a standard deviation of 10, previous studies have suggested 2.0–5.0 points as the MIC threshold in other populations (21, 30, 31). Evaluation of the PROMIS baseline data revealed that 5% change in baseline scores would range from 1.9-2.8 points for each of the 10 PROMIS measures. Thus, we set the MIC for all PROMIS measures at 2.5 points. For the HIT-6, a 5% change from baseline was estimated at 2.3 points, while the HIT-6 MIC in the literature is estimated at 1.5–2.5 (32). Therefore, we set the MIC for the HIT-6 at 2.5 points. A 5% change from baseline for the HCV-PRO was 3.6 points, thus we set the MIC for the HCV-PRO at 4 points. A 5% MIC change in the TMSASx10 score was 3.0 points. These MIC thresholds are conservative estimates to mitigate the risk of committing Type I error (i.e., false positives). A PRO change score that increased >5% is suggestive of clinically significant worse symptoms, whereas a PRO change score that decreased >5% is suggestive of clinically significant improvements.

Multivariable regression models using data-splitting strategies.—Generalized linear regression models were used for both exploratory and confirmatory evaluation of predictors of PRO change from baseline to on-treatment and baseline to post-treatment. Absolute mean PRO change scores (continuous) were used as the dependent variables. We used a two-stage modeling strategy based on data splitting. Participants were randomly assigned to two groups: Sample 1 or Sample 2.

Sample 1 was used for exploratory model building efforts to generate a set of candidate predictor variables that might be associated with change in each PRO. Model building with Sample 1 relied on unsupervised use of least absolute shrinkage and selection operator (LASSO) methods and model averaging algorithms. Sample 2 was used for confirmation. In Sample 2, the variables in the hypothesized model were considered validated if their regression coefficients were statistically significant at p<0.01. Candidate predictor variables available for inclusion in the exploratory models in Sample 1 included the following baseline covariates: age, sex, race, education, income, employment, cirrhosis status, alcohol abuse, substance use, and mental health issues, ethnicity, MELD score in cirrhosis patients, HIV, DAA treatment cohort, ribavirin (RBV) use, treatment duration, treatment experience, and number of health comorbidities.

Multiple imputation.—For use in multivariable regression models, missing values of baseline covariates (not PROs) were assumed to satisfy the missing-at-random criterion and were addressed via multiple imputations. A multivariate multiple imputation algorithm (SAS procedure MI) was used to generate 20 completed copies of the dataset. Each statistical regression model of interest was fitted to all 20 datasets. The 20 sets of results were combined (SAS procedure MIANALYZE) to produce the final results for each multivariable regression model.

Sensitivity Analyses.—To guide our level of trust in the main results, sensitivity analyses were performed in which the methods and assumptions used were perturbed using variations on multiple age (years) versus using age-groups), use of alternative methods for addressing confounding, and imputation variable-selection methods (with and without supervision), definitions of variables (e.g., exploring the two on-treatment assessment windows (T2/T3) separately to compare with our main results.

Statistical Computations.—Statistical computations were performed using SAS System software version 9.4 (SAS Institute, Cary, NC). PROMIS T-scores were computed using R software, version 3.1.2 (2014 The R Foundation for Statistical Computing), and RStudio software, version 1.0.136 (RStudio Inc.).

Results

Study Flowchart

The study flowchart is provided in Figure 1. Of the 1601 patients enrolled, 1564 patients (98%) completed PROs early on-treatment (T2), late on-treatment (T3), or at 12-weeks post-treatment (T4) and were included in the analyses of the total cohort. The cohort of patients prescribed daclatasvir/sofusbuvir were excluded due to low sample size (n=22). Of those with baseline and post-treatment PROs, 1410 patients (90%) had post-treatment HCV RNA available to determine SVR status. Of these, 1346 (95%) achieved SVR. A total of 154 patients had missing HCV RNA data: 10 died, 4 withdrew before T4 and 140 did not return for post-treatment follow-up labs.

Patient characteristics

Table 1 describes the baseline characteristics of the total cohort and the four DAA subgroups. The majority of patients were prescribed SOF/LED (63%) and 5% were prescribed PrOD. In the overall cohort 55% were male, 33% were Black (61% White), and mean age 58 years (SD=11; range 23–86). The majority were from lower socioeconomic status groups with 54% high school degree, 74% < \$40,000 per year income, and 45% receiving or applying for disability benefits. Over half of patients had 4+ health comorbidities, with an average of 4 (range: 0–17). The majority were infected with HCV genotype 1, 4, or 6 (83%), 82% received 12 weeks of therapy and 13% were prescribed RBV. Of the 47% classified as having cirrhosis, 14% had a MELD 12. Notably, several other patient subgroups often under-represented in registration trials were well-represented in this cohort: 39% non-white, 37% with mental health issues, 15% with alcohol abuse, and 23% with substance use.

Patient characteristics across the four DAA subgroups were similar except for the cohort prescribed GRZ/ELB, which included a higher proportion of patients who were black, on disability, with lower income, with a higher number of comorbidities, and higher mean creatinine scores, likely contributing to elevated MELD scores. As expected, more patients on PrOD were prescribed RBV.

Change in PRO scores from baseline to on- and post-treatment

Figure 2 shows the mean change and MIC threshold value for 13 PROs from baseline to T2, T3 and T4 in the total cohort (n=1,564). During treatment, the majority of PRO scores reduced slightly from baseline (- sign suggests symptom reduction); however, diarrhea, nausea, and headache increased slightly early in treatment (+ sign suggest symptom worsens). The magnitude of PRO mean changes during DAA therapy was small and none of the mean changes reached the 5% clinically significant MIC thresholds. All PRO means improved from baseline to post-treatment (-signs suggest improvements) with fatigue, sleep disturbance and functional well-being reaching the MIC thresholds for >5% clinically significant improvements.

We also evaluated PRO mean change scores stratified by four DAA subgroups (Supplemental Materials Figs. 1a–1c). Overall, the magnitude of PRO mean changes was very small and the *vast majority* of changes did *not* reach the 5% clinical thresholds. However, some trends are worth noting. Whereas the majority of PRO mean change scores reduced slightly during treatment for patients prescribed SOF/LED, SOF/VEL and GRZ/ ELB, patients prescribed PrOD consistently showed worse PRO change scores during treatment, with diarrhea and nausea mean change scores reaching the MIC for clinically worse symptoms. Finally, the only clinically significant improvement in PRO on-treatment was a 6.5 point improvement in HCV-PRO functional well-being in the GRZ/ELB cohort.

Proportion of change in PROs and pre-existing conditions from baseline to on-treatment

While the *mean* score changes in the overall cohort from baseline to on-treatment were small in magnitude, we observed wide variability in individual patient's change scores. Figure 3 shows the proportion of patients deemed to have little or no change (<5% change from baseline in either direction), those with clinically significant *improvements*, and those with clinically significant *worsening*. For example, over 40% reported substantial improvements in fatigue, sleep disturbance, overall symptom burden and functional well-being, approximately 30% experienced worsening of these symptoms.

PRO changes stratified by SVR status

As previously noted, of 1,410 patients with evaluable SVR data, 1,346 (95%) achieved SVR and 64 (5%) did not, with 140 (10%) patients lost to follow-up with missing post-treatment HCV RNA data. As shown in Figure 4, all PRO mean scores improved from baseline to early post-treatment in patients who achieved SVR, with clinically significant improvements in fatigue (-4.1), sleep disturbance (-3.0) and functional well-being (-6.4), and a trend for depression and pain to improve (-2.3). Among the 64 patients who did not achieve SVR, five mean PRO scores worsened during early post-treatment and no PROs showed clinically significant improvements (>5% MIC), with the exception of fatigue (-2.8). Similarly, Figure

5 shows the percent change in PRO mean scores from baseline to early post-treatment. Patients who achieved SVR exhibited the most clinically significant improvements in functional well-being (20%), overall symptom burden (13%), and somatic symptoms. In contrast, the 64 patients who did not achieve SVR had worsening or negligible changes for most PROs except a few. It is essential to note that 91% of all patients who completed their T4 post-treatment surveys were unaware of their SVR status at the time of survey completion.

Page 9

Compared to patients who achieved SVR, the 64 patients who did not achieve SVR were more likely to be male (62% vs.55%), had cirrhosis (56% vs. 47%), treatment experience (33% vs. 18%), longer duration of therapy (13% vs. 8%), RBV use (19% vs. 13%) and APRI >2.0 (20% vs. 14%) (Table 2).

Compared to patients who achieved SVR, those who did not have a follow-up HCV RNA test (n=140) tended to be younger (15% vs. 5%), have lower income (89% vs. 73%), higher rates of unemployment (17% vs. 6%), mental health issues (54% vs. 35%), and treatment inexperience (91% vs. 82%) but they *did not* have a higher rate of alcohol abuse (16% vs. 15%) or drug use (24% vs. 22%) (Table 2).

Multivariable models for symptom changes on- and post-treatment

Using data-splitting strategies and data from all patients (n=1564), multivariable analyses showed that ages 35 to 55 (for anger and overall symptom burden) and baseline mental health issues (for depression, fatigue, functional well-being) were the most consistent independent predictors of symptom improvements during DAA therapy (Suppl Table 1). We also found that patients prescribed PrOD had worse of overall symptom burden (1.8 [0.4, 3.2, p<0.01]) and a trend towards worse fatigue (3.3 [-0.1,6.8]; p=0.06) during therapy relative to patients prescribed SOF/LED. Other patient characteristics predictive of greater symptom improvements included other race, being disabled, having a higher number of comorbidities, and being treatment naive.

Similarly, multivariable analyses of patients who achieved SVR (n=1,346) showed that ages 35 to 55 was the most consistent independent predictor of post-treatment symptom improvements including anger, anxiety, fatigue, abdominal pain, and overall symptom burden (Suppl Table 2). Other patient characteristics associated with greater symptom improvements after viral cure included age 20–34, white race, a higher number of health comorbidities, mental health disturbance, and substance use at baseline. Patients prescribed PRoD experienced less improvements in anxiety after treatment compared to patients treated with SOF/LED. The majority (71%) of the PrOD cohort also received RBV compared to 12% in other DAA cohorts.

It is important to note that the following patient- and treatment-level characteristics were not selected as predictors in any of the final multivariable models: sex, ethnicity, cirrhosis status, MELD, HIV, income, education, alcohol use, or treatment duration.

Sensitivity Analysis

Several sensitivity analyses were performed in which the methods and assumptions used were perturbed using variations on multiple imputation, variable-selection methods, definitions of variables, and alternative methods for addressing confounding. We also explored the two on-treatment assessment windows (T2/T3) separately to compare with our main results. These and all analyses produced results similar to the main analysis described above, increasing trust in the main results.

Discussion

Many patients with chronic HCV experience symptoms that may be attributed to their disease (2, 3). These patients look forward to viral eradication that may ameliorate those troublesome symptoms. Although interferon-free DAA regimens have been shown to be well tolerated in clinical trials, these reports require confirmation from a broader and more heterogeneous cohort of patients treated in real world clinical settings. The current study represents the largest, most comprehensive investigation of patients' experiences during and after treatment with several interferon-free DAA regimens prescribed in clinical practice. This real-world clinical cohort was geographically heterogeneous and diverse with regard to race, cirrhosis status, and a wide range of comorbidities, including psychiatric, alcohol and substance use issues (12, 13). To our knowledge, this is the first study to provide comparative PRO data collected from patients prescribed DAA regimens developed by different pharmaceutical companies and to include PRO data from patients who did and did not achieve viral cure. Finally, we determined whether these PRO changes were clinically meaningful using a conservative threshold. It is important to note that the vast majority (91%) of patients had no knowledge of SVR status before completing their post-treatment PRO surveys, providing strong evidence that improvements in PROs post-treatment are likely the result of viral eradication on biological processes, and not mere psychological placebo effect.

An important observation in this study is that while the average change in PRO scores of the total cohort was small, one-quarter to one-third reported *worsening* of symptoms during DAA therapy. Previous PRO studies focused on reporting overall mean change scores, but neglected to describe the full distribution of patients' experiences. Our findings have implications for how clinicians might help set expectations with patients initiating DAA therapy, for example providing a balanced perspective that some patients experience no changes, others experience improvements, but a third may experience worsening of baseline symptoms. In accordance with other studies, PRO scores improved from baseline to post-treatment, with clinically meaningful improvements specifically in fatigue, sleep disturbance and functional well-being. These improvements were more pronounced in patients who achieved SVR (5), our study uniquely demonstrates that sleep and pain issues improve after viral eradication. This is particularly meaningful since the prevalence of sleep and pain disorders are high among patients with chronic HCV (2, 33–36).

We observed that younger patients (ages 35 to 55), and those with mental health issues or more health comorbidities reported the greatest symptom improvements during and after

therapy. These patients represent vulnerable subgroups with psychosocial and medical challenges who may benefit the most from being given the opportunity to rid themselves of a stigmatizing infectious disease and engage in healthcare. These patients tended to have the worst symptoms at baseline with more room for improvement (data not shown), suggesting the possibility of regression towards the mean. We found that cirrhosis was not independently associated with PRO changes during therapy or after viral cure indicating that patients with and without cirrhosis tend to experience similar symptom benefits. It should be clarified that most patients with cirrhosis in this cohort had compensated liver disease.

Overall, PRO changes in the cohorts of patients prescribed SOF/LED, SOF/VEL or GRZ/ELB were similar across the DAA cohorts while on-treatment with overall improvement in symptoms after treatment completion, 33% of which were clinically substantial improvements. Therefore, patients can be counseled regarding the overall stability of symptoms during HCV therapy and improvements post-SVR regardless of which of these regimens are prescribed. Patients prescribed PrOD had less improvement in PROs during treatment, including clinically significant diarrhea and nausea, and experienced the least improvements after completing treatment. It should be noted that the number of patients who received PrOD was small and a higher proportion received ribavirin, which likely contributed to worsening symptoms during treatment. Nevertheless, PrOD is no longer used in the U.S. and many other countries, having been superseded by simpler, better tolerated DAA regimens.

In this real-world cohort, 9% of patients did not return for follow-up HCV RNA testing to determine SVR status. This was an unexpected finding since one would assume that most patients understood that DAAs are expensive and many had to wait a long time for insurance approval in order to receive treatment. Patients who never came back for follow-up HCV RNA testing were disproportionately younger, had lower incomes, higher rates of unemployment, disability, mental health issues, and were more likely to be treatment naïve. Contrary to speculation, these patients who were non-compliant with follow-up did *not* have higher rates of baseline drug or alcohol abuse. These data indicate the importance of educating all patients about the importance of post-treatment laboratory tests, in particular younger patients and those at greater risk for being lost to clinical follow-up.

A few limitations are worth noting. The observational study design precludes any definitive head-to-head statistical comparisons of DAA regimens. Amongst DAA regimens, SOF/LED was the most commonly prescribed while the other three DAA regimens were prescribed less often. This study also included a relatively small number of patients with advanced liver disease or liver transplantation. Aside from laboratory tests, minimal clinical data were extracted from electronic health records. The scope of this study did not include clinical data on fibrosis staging, decompensation events, medications or comorbid liver diseases or chronic illnesses. Thus, by design, we have not compared patient-reported information with clinical health data or to classify cirrhosis according to Child-Pugh scores. Our findings may not generalize to other subpopulations or clinical settings, including younger people who are actively injecting drugs or persons receiving medication-assisted treatment for opioid use disorders.

The strengths of this study are worth noting. This study is the largest comprehensive realworld PRO study during and after treatment with different DAA regimens. We provided comparative data on four commonly prescribed DAA regimens. The study population included many subpopulations under-represented in registration trials. Unlike prior studies, we provided the full spectrum of patients' experiences including those who experienced worsening of symptoms. Although the non-SVR subgroup was small (n=64), our study suggests that there may be differences in changes in PROs post-treatment between patients who do and do not achieve SVR. These data are consistent with the positive effects of viral eradication on patients' functioning and specific symptoms. Finally, PROP UP has been a highly patient-centered study since its inception with patients engaged throughout all phases of study development to ensure that our findings are meaningful and relevant to people affected by the disease (12, 39).

CONCLUSION

This comprehensive assessment of changes in neuropsychiatric, somatic and gastrointestinal symptoms, and functional well-being during and after therapy with all-oral DAA therapies provides new insights relevant to patients, clinicians and other stakeholders.

Supplementary Material

Refer to Web version on PubMed Central for supplementary material.

Acknowledgements

The authors would like to acknowledge the contributions of the following people: Alan Franciscus of www.HCVadvocate.org, Anquenette Sloan, Summer Wadsworth-Delciotto, Scott Kixmiller, Larry Huston, Finton Brown (UNC Patient Engagement Group); Virginia Sharpless, Ken Berquist, Herleesha Anderson, Courtenay Pierce, Jenn Barr, Bryonna Jackson, Jane Giang, Jama Darling, Paul Hayashi, Steven Zacks, A. Sidney Barritt IV, Scott Elliot, Dawn Harrison, Danielle Cardona (University of North Carolina); Patrick Horne (University of Florida); Kelly Borges, Danielle Ciuffetelli (University of Pennsylvania); Chrissy Ammons, Kathleen Genther, Jessica Mason (Virginia Commonwealth University); Lelani Fetrow, Vicki Shah (Rush University); Theresa Cattoor, Alisha McLendon (Saint Louis University); Mariechristi Candido, Sophia Zaragoza, Sandeep Dhaliwal, Patricia Poole, Rebecca Hluahanich, Kathleen Haight, (University of California, Davis); Andrea Gajos, Elizabeth Wu, Carrie Bergmans (University) of Michigan); AnnMarie Liapakis, MD, Kristine Drozd, Carol Eggers, Hong Chau and Claudia Bertuccio (Yale University). William Harlan, Roberta Golden, Kylee Diaz(Asheville Gastroenterology Associates), William King, Megan Marles (Wilmington Gastroenterology Associates). We also extend our gratitude to the PROP UP study participants. We would like to acknowledge the following grants: PCORI CER-1408-20660 (Evon); NIH P30-DK34987 (Sandler); NCATS UL1TR002489; NIH K24-HD06920 (Golin); P30 AI-50410 (Golin).

Financial support: Research reported in this publication was funded through a Patient-Centered Outcomes Research Institute (PCORI) Award to Donna Evon (CER-1408-20660). Additional support for this study (data management) came from the NIDDK-funded Center for Gastrointestinal Biology and Disease (CGIBD; PI: Sandler; P30-DK34987) and the National Center for Advancing Translational Sciences (NCATS), National Institutes of Health Grant Number UL1TR002489. Additional support for Dr. Golin's salary was partially supported by the Eunice Kennedy Shriver National Institute of Child Health and Human Development (K24-HD06920) and by the University of North Carolina Center for AIDS Research (CFAR) (P30 AI-50410). The statements in this publication are solely the responsibility of the authors and do not necessarily represent the views of PCORI, its Board of Governors or Methodology Committee, or the NIH.(37,38)

Selected abbreviations

ALT

alanine aminotransferase test

AST	aspartate aminotransferase test
APRI	AST to platelet ratio index
AUDIT	Alcohol Use Disorders Identification Test
CI	Confidence interval
DAA	Direct acting antiviral
FIB-4	Fibrosis-4 Index for Liver Fibrosis
GI	Gastrointestinal
HCV	hepatitis C virus
MSAS	Memorial Symptom Assessment Scale
PCORI	Patient-Centered Outcomes Research Institute
PRO	patient-reported outcome
PROMIS®	Patient-Reported Outcome Measurement Information System®
PROP UP	The Patient-Reported Outcomes Project of HCV-TARGET
RBV	Ribavirin
SD	standard deviation
TMSAS	Total Memorial Symptom Assessment Scale

References

- Kleinman L, Mannix S, Yuan Y, Kummer S, L'Italien G, Revicki D. Review of patient-reported outcome measures in chronic hepatitis C. Health Qual.Life Outcomes. 2012;10:92. doi: 10.1186/1477-7525-10-92.:92–10. [PubMed: 22871087]
- Lang CA, Conrad S, Garrett L, Battistutta D, Cooksley WG, Dunne MP, Macdonald GA. Symptom prevalence and clustering of symptoms in people living with chronic hepatitis C infection. J.Pain Symptom.Manage. 2006;31:335–344. [PubMed: 16632081]
- Ferri C, Ramos-Casals M, Zignego AL, Arcaini L, Roccatello D, Antonelli A, Saadoun D, et al. International diagnostic guidelines for patients with HCV-related extrahepatic manifestations. A multidisciplinary expert statement. Autoimmun Rev 2016;15:1145–1160. [PubMed: 27640316]
- 4. Younossi ZM, Stepanova M, Zeuzem S, Dusheiko G, Esteban R, Hezode C, Reesink HW, et al. Patient-reported outcomes assessment in chronic hepatitis C treated with sofosbuvir and ribavirin: The VALENCE study. J Hepatol 2014;61:228–234. [PubMed: 24713186]
- Younossi Z, Henry L. Systematic review: patient-reported outcomes in chronic hepatitis C the impact of liver disease and new treatment regimens. Aliment Pharmacol Ther 2015;41:497–520. [PubMed: 25616122]
- Younossi ZM, Stepanova M, Feld J, Zeuzem S, Jacobson I, Agarwal K, Hezode C, et al. Sofosbuvir/ velpatasvir improves patient-reported outcomes in HCV patients: Results from ASTRAL-1 placebocontrolled trial. J Hepatol 2016;65:33–39. [PubMed: 26956698]
- 7. Bero L. Industry sponsorship and research outcome: a Cochrane review. JAMA Intern Med 2013;173:580–581. [PubMed: 23440226]

- Lundh A, Lexchin J, Mintzes B, Schroll JB, Bero L. Industry sponsorship and research outcome. Cochrane Database Syst Rev 2017.
- 9. Zepatier prescription information: http://www.merck.com/product/usa/pi_circulars/z/zepatier/ zepatier_pi.pdf.
- Epclusa prescription information: http://www.gilead.com/~/media/Files/pdfs/medicines/liverdisease/epclusa/epclusa_pi.pdf.
- 11. Harvoni prescription information: https://www.gilead.com/~/media/Files/pdfs/medicines/liverdisease/harvoni/harvoni_pi.pdf
- Evon DM, Golin CE, Stewart P, Fried MW, Alston S, Reeve B, Lok AS, et al. Patient engagement and study design of PROP UP: A multi-site patient-centered prospective observational study of patients undergoing hepatitis C treatment. Contemp Clin Trials 2017;57:58–68. [PubMed: 28342989]
- 13. Evon DM, Stewart PW, Amador J, Serper M, Lok AS, Sterling RK, Sarkar S, et al. A comprehensive assessment of patient reported symptom burden, medical comorbidities, and functional well being in patients initiating direct acting antiviral therapy for chronic hepatitis C: Results from a large US multi-center observational study. PLoS One, 8 1;13(8):e0196908. doi: 10.1371/journal.pone.0196908. eCollection 2018 [PubMed: 30067745]
- Pence BW, Gaynes BN, Whetten K, Eron JJ Jr., Ryder RW, Miller WC. Validation of a brief screening instrument for substance abuse and mental illness in HIV-positive patients. J.Acquir.Immune.Defic.Syndr 2005;40:434–444. [PubMed: 16280698]
- Saunders JB, Aasland OG, Babor TF, de IF Jr., Grant M. Development of the Alcohol Use Disorders Identification Test (AUDIT): WHO Collaborative Project on Early Detection of Persons with Harmful Alcohol Consumption--II. Addiction 1993;88:791–804. [PubMed: 8329970]
- Wai CT, Greenson JK, Fontana RJ, Kalbfleisch JD, Marrero JA, Conjeevaram HS, Lok AS. A simple noninvasive index can predict both significant fibrosis and cirrhosis in patients with chronic hepatitis C. Hepatology 2003;38:518–526. [PubMed: 12883497]
- 17. United Network for Organ Sharing (UNOS). MELD/PELD Calculator Documentation. https://optn.transplant.hrsa.gov/resources/allocation-calculators/meld-calculator/.
- Evon DM, Amador J, Stewart P, Reeve BB, Lok AS, Sterling RK, Di Bisceglie AM, et al. Psychometric properties of the PROMIS short form measures in a U.S. cohort of 961 patients with chronic hepatitis C prescribed direct acting antiviral therapy. Aliment Pharmacol Ther 2018, 4;47(7):1001–1011. [PubMed: 29377191]
- Cella D, Riley W, Stone A, Rothrock N, Reeve B, Yount S, Amtmann D, et al. The Patient-Reported Outcomes Measurement Information System (PROMIS) developed and tested its first wave of adult self-reported health outcome item banks: 2005–2008. J Clin Epidemiol 2010;63:1179–1194. [PubMed: 20685078]
- Reeve BB, Hays RD, Bjorner JB, Cook KF, Crane PK, Teresi JA, Thissen D, et al. Psychometric evaluation and calibration of health-related quality of life item banks: plans for the Patient-Reported Outcomes Measurement Information System (PROMIS). Med.Care. 2007;45:S22–S31. [PubMed: 17443115]
- 21. Khanna D, Hays RD, Shreiner AB, Melmed GY, Chang L, Khanna PP, Bolus R, et al. Responsiveness to Change and Minimally Important Differences of the Patient-Reported Outcomes Measurement Information System Gastrointestinal Symptoms Scales. Dig Dis Sci 2017;62:1186–1192. [PubMed: 28251500]
- Kosinski M, Bayliss MS, Bjorner JB, Ware JE Jr., Garber WH, Batenhorst A, Cady R, et al. A sixitem short-form survey for measuring headache impact: the HIT-6. Qual Life Res 2003;12:963– 974. [PubMed: 14651415]
- 23. Everson GT, Towner WJ, Davis MN, Wyles DL, Nahass RG, Thuluvath PJ, Etzkorn K, et al. Sofosbuvir With Velpatasvir in Treatment-Naive Noncirrhotic Patients With Genotype 1 to 6 Hepatitis C Virus Infection: A Randomized Trial. Ann Intern Med 2015;163:818–826. [PubMed: 26551051]
- 24. Bajaj JS, Thacker LR, Wade JB, Sanyal AJ, Heuman DM, Sterling RK, Gibson DP, et al. PROMIS computerised adaptive tests are dynamic instruments to measure health-related quality of life in patients with cirrhosis. Aliment Pharmacol Ther 2011;34:1123–1132. [PubMed: 21929591]

- 25. Portenoy RK, Thaler HT, Kornblith AB, Lepore JM, Friedlander-Klar H, Kiyasu E, Sobel K, et al. The Memorial Symptom Assessment Scale: an instrument for the evaluation of symptom prevalence, characteristics and distress. Eur J Cancer 1994;30a:1326–1336. [PubMed: 7999421]
- Chang VT, Hwang SS, Thaler HT, Kasimis BS, Portenoy RK. Memorial symptom assessment scale. Expert Rev Pharmacoecon Outcomes Res 2004;4:171–178. [PubMed: 19807521]
- 27. Anderson RT, Baran RW, Dietz B, Kallwitz E, Erickson P, Revicki DA. Development and initial psychometric evaluation of the hepatitis C virus-patient-reported outcomes (HCV-PRO) instrument. Qual Life Res 2014;23:561–570. [PubMed: 24005885]
- 28. Anderson RT, Baran RW, Erickson P, Revicki DA, Dietz B, Gooch K. Psychometric evaluation of the hepatitis C virus patient-reported outcomes (HCV-PRO) instrument: validity, responsiveness, and identification of the minimally important difference in a phase 2 clinical trial. Qual Life Res 2014;23:877–886. [PubMed: 24036634]
- 29. Page P. Beyond statistical significance: clinical interpretation of rehabilitation research literature. Int J Sports Phys Ther 2014;9:726–736. [PubMed: 25328834]
- Amtmann D, Kim J, Chung H, Askew RL, Park R, Cook KF. Minimally important differences for Patient Reported Outcomes Measurement Information System pain interference for individuals with back pain. J Pain Res 2016;9:251–255. [PubMed: 27175093]
- Hays RD, Spritzer KL, Fries JF, Krishnan E. Responsiveness and minimally important difference for the patient-reported outcomes measurement information system (PROMIS) 20-item physical functioning short form in a prospective observational study of rheumatoid arthritis. Ann Rheum Dis 2015;74:104–107. [PubMed: 24095937]
- 32. Smelt AF, Assendelft WJ, Terwee CB, Ferrari MD, Blom JW. What is a clinically relevant change on the HIT-6 questionnaire? An estimation in a primary-care population of migraine patients. Cephalalgia 2014;34:29–36. [PubMed: 23843470]
- Hussain KB, Fontana RJ, Moyer CA, Su GL, Sneed-Pee N, Lok AS. Comorbid illness is an important determinant of health-related quality of life in patients with chronic hepatitis C. Am J Gastroenterol 2001;96:2737–2744. [PubMed: 11569704]
- Morasco BJ, Huckans M, Loftis JM, Woodhouse J, Seelye A, Turk DC, Hauser P. Predictors of pain intensity and pain functioning in patients with the hepatitis C virus. Gen.Hosp.Psychiatry. 2010;32:413–418. [PubMed: 20633746]
- 35. Rivera J, de Diego A, Trinchet M, Garcia Monforte A. Fibromyalgia-associated hepatitis C virus infection. Br J Rheumatol 1997;36:981–985. [PubMed: 9376995]
- Barkhuizen A, Rosen HR, Wolf S, Flora K, Benner K, Bennett RM. Musculoskeletal pain and fatigue are associated with chronic hepatitis C: a report of 239 hepatology clinic patients. Am J Gastroenterol 1999;94:1355–1360. [PubMed: 10235218]
- 37. Hsu YC, Lin JT, Ho HJ, Kao YH, Huang YT, Hsiao NW, Wu MS, et al. Antiviral treatment for hepatitis C virus infection is associated with improved renal and cardiovascular outcomes in diabetic patients. Hepatology 2014;59:1293–1302. [PubMed: 24122848]
- Mahale P, Engels EA, Li R, Torres HA, Hwang LY, Brown EL, Kramer JR. The effect of sustained virological response on the risk of extrahepatic manifestations of hepatitis C virus infection. Gut 2018;67:553–561. [PubMed: 28634198]
- Evon DM, Golin CE, Stoica T, Jones RE, Willis SJ, Galanko J, Fried MW. What's Important to the Patient? Informational Needs of Patients Making Decisions About Hepatitis C Treatment. Patient 2017 Jun;10(3):335–344.

Highlights

- Overall change in symptoms and functioning on DAAs was not clinically meaningful
- Patients' experiences are very heterogeneous
- Patients prescribed one DAA regimen experienced the worst symptoms
- Patients cured have clinical improvement in fatigue, sleep, and functioning
- Patients not cured have minimal improvements



Fig. 1. Study flowchart

NOTE: DAC/SOF: daclatasvir/sofosbuvir, T1: baseline, T2: early on-treatment, T3: late on-treatment, T4: early post-treatment.

Evon et al.



Fig. 2. Mean PRO change scores at on-treatment and post-treatment in overall sample (n=1564) NOTE: Unadjusted PRO change scores at the following time points are shown: Early On-Tx: Early Treatment Phase; Late On-Tx: Late Treatment Phase; Post-Tx: Early Post-Treatment. DAA: Direct-Acting Antiviral. MIC: Minimally Important Change defined as > 5% change in PRO score suggests clinically significant change. The 5% MIC threshold for the PROMIS and Headache measures is ± 2.5 points; the MIC for Functional Well-Being is ± 4 points; the MIC for Overall Symptom Burden is ± 3.0 . Negative change scores=PRO score improved; Positive change scores=PRO score worsened. Missing values for all PROs were 4%–8% (functional well-being missing 14%–16%).

Evon et al.



Fig. 3. Proportion of patients whose symptoms stayed the same, improved or worsened during DAA therapy

NOTE: DAA: Direct-Acting Antiviral. Missing values for all PROs were 1%–3%, except Functional well-being was missing for 9% of patients. Improved 5% improvement from baseline; Worsened 5% worse from baseline score.

PRO measure		Mean change score	SD	95% CI for the mean change score
Depression S	SVR SVR	-2.3 0.2	9 8	
Anger S	SVR	-2.1	11	
Non-S	SVR	0.7	10	
Anxiety S	SVR	-1.6	9	H O I
Non-S	SVR	-0.8	9	
Cognitive concerns S Non-S	SVR SVR	-1.3 -0.1	8 7	
Pain interference S	SVR	-2.3	9	H e H
Non-S	SVR	-1.8	9	
Fatigue S	SVR	-4.1	10	⊢● ↓
Non-S	SVR	-2.8	10	▶──●
Sleep disturbance SNon-S	SVR SVR	-3.0 -0.7	10 9	
Headache	SVR SVR	-1.5 -1.3	8 7	H O I
Abdominal pain S	SVR	-2.2	13	
Non-S	SVR	0.9	11	
Diarrhea S	SVR	-0.1	10	⊢
Non-S	SVR	0.4	11	⊢
Nausea S	SVR	-1.4	8	
Non-S	SVR	-0.2	7	
Overall treatment S	SVR	-1.8	4	₽
burden Non-S	SVR	-1.2	4	
Functional S	SVR	-6.4	17	
well-being Non-S	SVR	2.0	19	
			- E	8 -4 0 4 8 Better

Fig. 4. PRO mean change scores from baseline to early post-treatment by SVR status NOTE: PRO mean change scores from Baseline to Early Post-Treatment (Early Post-Tx). CI: confidence interval; SVR: Sustained Virologic Response. For patients who achieved SVR, functional well-being change scores were missing for 13%–22% of patients; for the ten PRO measures, change scores were missing for 0% - 6% of patients. Among non-SVR patients, for all PRO measures, change scores were missing for 0% - 17% of patients.





NOTE: Percent change in PRO mean scores from Baseline to Early Post-Treatment (Early Post-Tx). Horizontal bars denotes 95% confidence intervals. *The lower limit of the 95% CI = -42 for Overall Symptom Burden in the Non-SVR group. Dotted vertical lines represent 5% MIC thresholds. SVR: Sustained Virologic Response. For patients who achieved SVR (n=1346), functional well-being change scores were missing for 15% of patients; for the ten PRO measures, change scores were missing for 0% - 5% of patients. For patients who did not achieve SVR (n=64), headache and functional well-being change scores were missing for 17% and 20% of patients, respectively; for the eight other PRO measures, change scores were missing for 0% - 13% of patients.

Table 1.

Patient characteristics by DAA therapy cohort

	DAA cohort therapy					
Characteristics	А	s	S	G	Р	
		OF/LE	OF/VE	RZ/EL		
	п	D	L	В	rOD	
	(n=1564)	(n=989)	(n=335)	(n=170)	(n=70)	
			n (%)			
Sociodemographic features						
	5	5	5	5	5	
Age, years (mean, SD)	8 (11)	8 (10)	7 (11)	9 (10)	4 (12)	
<35	8	5	1	7	9	
	6 (5)	2 (5)	8 (5)	(4)	(13)	
25.55	3	2	8	4	1	
30-00	72 (24)	25 (23)	9 (27)	0 (24)	8 (26)	
. 55	1	7	2	1	4	
>>>	106 (71)	12 (72)	28 (68)	23 (72)	3 (61)	
Sex						
M 1.	8	5	1	9	3	
Male	67 (55)	45 (55)	93 (58)	2 (54)	7 (53)	
	6	4	1	7	3	
Female	97 (45)	44 (45)	42 (42)	8 (46)	3 (47)	
Race						
	5	3	4	8	1	
ВІаск	12 (33)	61 (37)	9 (15)	4 (50)	8 (26)	
XX 75 °.	9	5	2	7	4	
White	53 (61)	77 (59)	49 (75)	8 (46)	9 (70)	
0.1	9	4	3	7	3	
Other	4 (6)	8 (5)	6 (11)	(4)	(4)	
Ethnicity						
TT	6	3	2	8	1	
Hispanic/Latino	3 (4)	1 (3)	3 (7)	(5)	(1)	
Non-Hispanic/other	1	9	2	1	6	
	427 (96)	11 (97)	92 (93)	57 (95)	7 (99)	
Education						
	8	5	1	9	3	
High school diploma or equivalent	40 (54)	39 (55)	69 (51)	5 (56)	7 (53)	
	7	4	1	7	3	
> High school	09 (46)	42 (45)	60 (49)	4 (44)	3 (47)	
Annual household income						
- 0 40W	1	7	2	1	5	
< \$40K	134 (74)	09 (73)	32 (72)	39 (83)	2 (74)	

	DAA cohort therapy				
Characteristics	Α	S	S	G	Р
		OF/LE	OF/VE	RZ/EL	
	II	D	L	В	rOD
	(n=1564)	(n=989)	(n=335)	(n=170)	(n=70)
			n (%)		
Sociodemographic features					
\$40K	3	2	9	2	1
\$40K	95 (26)	58 (27)	0 (28)	9 (17)	8 (26)
Employment					
Working full or part time	5	3	1	3	3
working full of part time	40 (36)	40 (36)	37 (42)	1 (19)	2 (46)
Receiving/applying for disability	6	4	1	1	2
Receiving apprying for disability	76 (45)	37 (45)	15 (35)	05 (63)	9 (42)
Unemployed	1	6	3	1	4
Unemployed	08 (7)	0 (6)	4 (10)	0 (6)	(6)
Patirad/homomokar/student	1	1	3	2	4
Retried/homemakei/student	89 (12)	27 (13)	8 (12)	0 (12)	(6)
Liver Clinical, Laboratory and Treat	ment Marker	s			
Genotype					
Conotuno 1 4 or 6	1	9	7	1	7
Genotype 1, 4 of 0	287 (83)	70 (99)	9 (23)	68 (100)	0 (100)
Conotuno 2	1	3	1	0	0
Genotype 2	35 (9)	(0)	32 (40)	(0)	(0)
Conotuno 3	1	2	1	0	0
denotype 5	25 (8)	(0)	23 (37)	(0)	(0)
Cirrhosis status					
Cirrhotic	7	4	1	7	3
Chinote	38 (47)	58 (46)	68 (51)	7 (46)	5 (50)
Noncirrhotic	8	5	1	9	3
Witelifilotte	18 (53)	28 (54)	64 (49)	1 (54)	5 (50)
MELD status in cirrhotic patients					
MELD 6 11	5	3	1	4	2
	27 (86)	43 (89)	17 (83)	4 (68)	3 (92)
MELD 12	8	4	2	2	2
MEED 12	9 (14)	2 (11)	4 (17)	1 (32)	(8)
AST to Platelet Ratio Index					
(APRI)					
APRI 20	1	8	2	1	6
111 NI 2.0	293 (86)	21 (87)	52 (79)	58 (93)	2 (91)
$\Delta PRI > 2.0$	2	1	6	1	6
111 NI > 2.0	08 (14)	22 (13)	9 (21)	1 (7)	(9)
ALT, U/L (mean (SD))	7	7	8	5	7

	DAA cohort therapy				
Characteristics	Α	s	s	G	Р
		OF/LE	OF/VE	RZ/EL	
	II	D	L	В	rOD
	(n=1564)	(n=989)	(n=335)	(n=170)	(n=70)
			n (%)		
Sociodemographic features					
	8 (69)	8 (69)	9 (78)	7 (48)	8 (63)
Creatinine mg/dI (mean (SD))	1	1	1	2	1
Creatinine, ing/dE (incan (5D))	(1)	(0)	(0)	(3)	(1)
Health comorbidities					
0–1	3	1	8	1	1
	16 (20)	99 (20)	1 (24)	9 (11)	7 (24)
2-3	3	2	9	3	2
	94 (25)	47 (25)	4 (28)	2 (19)	1 (30)
4	8	5	1	1	3
	52 (55)	41 (55)	60 (48)	19 (70)	2 (46)
Current kidney disease					
No	1	9	3	1	6
	464 (94)	54 (97)	16 (96)	26 (75)	8 (99)
Yes	8	3	1	4	1
	8 (6)	2 (3)	3 (4)	3 (25)	(1)
Prescribed treatment duration					
8 weeks	1	1	3	1	0
	54 (10)	53 (16)	(1)	(1)	(0)
12 weeks	1	7	3	1	6
	286 (82)	52 (76)	15 (93)	58 (92)	1 (87)
16 or 24 weeks	1	8	2	1	9
	24 (8)	4 (8)	0 (6)	1 (7)	(13)
Ribavirin					
Without Ribavirin	1	8	3	1	2
	363 (87)	71 (88)	17 (95)	55 (91)	0 (29)
With Ribavirin	2	1	1	1	5
m , , , ,	01 (13)	18 (12)	8 (5)	5 (9)	0(71)
Treatment experience	1	7	2	1	~
reatment naive	1	7	2		6
Tractment experies and	252 (82)	90 (82)	00 (81)	33 (81) 2	3 (93) E
meannent experienced	Z 70 (10)	1	0 3 (10)	3 2 (10)	כ ריי
SVD Ashiovad	/0(10)	/0 (18)	3 (19)	2 (19)	()
Sy K Acmevea	C	2	1	0	А
110	0	3 2 (4)	1 0 (6)	9	4
Vac	4 (3)	∠ (4) Q	2 (U) 2	(0)	(/)
103	1	0	4	1	5

Yes

	DAA cohort therapy					
Characteristics	Α	S	S	G	Р	
		OF/LE	OF/VE	RZ/EL		
	II	D	L	В	rOD	
	(n=1564)	(n=989)	(n=335)	(n=170)	(n=70)	
			n (%)			
Sociodemographic features						
	346 (95)	66 (96)	79 (94)	44 (94)	7 (93)	
Mental Health and Substance Use	Features					
Mental health disturbance						
NT.	9	6	2	1	3	
NO	82 (63)	22 (63)	08 (62)	16 (69)	6 (51)	
X7	5	3	1	5	3	
Yes	76 (37)	64 (37)	25 (38)	3 (31)	4 (49)	
Alcohol abuse						
NT.	1	8	2	1	5	
NO	327 (85)	45 (86)	79 (84)	47 (87)	6 (80)	
V.	2	1	5	2	1	
Yes	29 (15)	38 (14)	5 (16)	2 (13)	4 (20)	
Substance use						
N	1	7	2	1	5	
INU	205 (77)	70 (78)	44 (73)	33 (78)	8 (83)	
	3	2	8	3	1	

 $52\,(23)$

15 (22)

NOTE: DAA: Direct-Acting Antiviral, SOF/LED: sofosbuvir/ledipasvir, SOF/VEL: sofosbuvir/velpatasvir, GRZ/ELB: grazoprevir/elbasvir, PrOD: paritaprevir/ombitasvir/ritonavir + dasabuvir, MELD: Model for End-Stage Liver Disease, AST: Aspartate Aminotransferase, ALT: Alanine Aminotransferase, SVR: Sustained Virologic Response. Missing values for all characteristics were 4%, except MELD and SVR were missing for 16%–29% and 10%–13% of patients, respectively.

8 (27)

7 (22)

2(17)

Table 2.

Patient characteristics stratified by SVR status

Characteristics	SVR(n=1346)	Non-SVR(n=64)	Lost to Follow-Up(n=140)			
	n (%)					
Sociodemographic features						
Age, years (mean (SD))	59 (10)	60 (8)	50 (12)			
<35	63 (5)	1 (2)	21 (15)			
35–55	293 (22)	15 (23)	62 (44)			
>55	990 (73)	48 (75)	57 (41)			
Sex						
Male	739 (55)	40 (62)	77 (55)			
Female	607 (45)	24 (38)	63 (45)			
Race						
Black	443 (33)	25 (39)	40 (29)			
White	814 (61)	37 (58)	94 (67)			
Other	84 (6)	2 (3)	6 (4)			
Education						
High school diploma or equivalent	721 (54)	32 (52)	79 (57)			
> High school	613 (46)	30 (48)	60 (43)			
Annual household income						
<\$40K	956 (73)	42 (69)	123 (89)			
\$40K	358 (27)	19 (31)	15 (11)			
Employment						
Working full or part time	480 (37)	23 (38)	35 (25)			
Receiving/applying for disability	567 (44)	30 (49)	69 (51)			
Unemployed	84 (6)	1 (2)	23 (17)			
Retired/homemaker/student	171 (13)	7 (11)	9 (7)			
Liver Clinical, Treatment and Labora	tory Markers					
Genotype						
Genotype 1, 4 or 6	1110 (83)	54 (84)	111 (79)			
Genotype 2	114 (9)	5 (8)	16 (12)			
Genotype 3	106 (8)	5 (8)	13 (9)			
Cirrhosis status						
Cirrhotic	626 (47)	36 (56)	67 (49)			
Noncirrhotic	716 (53)	28 (44)	69 (51)			
MELD status in cirrhotic patients						
MELD 6-11	451 (87)	29 (88)	45 (85)			
MELD 12	70 (13)	4 (12)	8 (15)			
AST to Platelet Ratio Index (APRI)						
APRI 2.0	1118 (86)	47 (80)	117 (87)			
APRI > 2.0	175 (14)	12 (20)	18 (13)			
ALT, U/L (mean (SD))	77 (68)	85 (68)	87 (82)			

Characteristics	SVR(n=1346)	Non-SVR(n=64)	Lost to Follow-Up(n=140)			
	n (%)					
Sociodemographic features						
Creatinine, mg/dL (mean(SD))	1 (1)	1 (0)	1 (1)			
Health comorbidities						
0–1	270 (20)	8 (13)	36 (26)			
2–3	326 (24)	22 (34)	46 (33)			
4	748 (56)	34 (53)	58 (41)			
Current kidney disease						
No	1256 (94)	60 (95)	137 (98)			
Yes	80 (6)	3 (5)	3 (2)			
Treatment experience						
Treatment naive	1099 (82)	43 (67)	110 (91)			
Treatment experienced	246 (18)	21 (33)	11 (9)			
Prescribed treatment duration						
8 weeks	139 (10)	5 (8)	9 (6)			
12 weeks	1101 (82)	51 (80)	123 (88)			
16 or 24 weeks	106 (8)	8 (13)	8 (6)			
Ribavirin						
Without Ribavirin	1171 (87)	52 (81)	127 (91)			
With Ribavirin	175 (13)	12 (19)	13 (9)			
Mental	Health and Subst	ance Use Features				
Mental health disturbance						
No	867 (65)	41 (64)	65 (46)			
Yes	473 (35)	23 (36)	75 (54)			
Alcohol abuse						
No	1144 (85)	52 (81)	117 (84)			
Yes	195 (15)	12 (19)	22 (16)			
Substance use						
No	1039 (78)	47 (73)	107 (76)			
Yes	300 (22)	17 (27)	33 (24)			

NOTE: DAA: Direct-Acting Antiviral, SVR: Sustained Virologic Response, MELD: Model for End-Stage Liver Disease, AST: Aspartate Aminotransferase, ALT: Alanine Aminotransferase. Missing values for all characteristics were 5%, except MELD was missing for 8%–21% of patients, and Treatment Experience was missing for 14% of the lost to follow-up patients.